

# The New Zealand Gazette

WELLINGTON: THURSDAY, 15 AUGUST 1991

### **Contents**

Parliamentary Summa	iry				 	 2654
Government Notices					 	 2660
Authorities and Other	Agenci	es of St	ate Not	ices	 	 2665
Land Notices					 	 2665
Regulation Summary					 	 2676

### Using the Gazette

The New Zealand Gazette, the official newspaper of the Government of New Zealand, is published weekly on Thursday. Publishing time is 4 p.m.

Notices for publication and related correspondence should be addressed to:

Gazette Office, Department of Internal Affairs, P.O. Box 805, Wellington. Telephone (04) 495 7200 Facsimile (04) 499 1865

or lodged at the Gazette Office, Seventh Floor, Dalmuir House, 114 The Terrace, Wellington.

Closing time for lodgment of notices at the Gazette Office is 12 noon on the Tuesday preceding publication (except for holiday periods when special advice of earlier closing times will be given).

Notices are accepted for publication in the next available issue, unless otherwise specified.

Notices being submitted for publication must be reproduced copies of the originals. Dates, proper names and signatures are to be shown clearly. A covering instruction setting out requirements must accompany all notices.

Copy will be returned unpublished if not submitted in accordance with these requirements.

#### **Availability**

The New Zealand Gazette is available on subscription from GP Publications Limited or over the counter from GP Books Limited bookshops at:

Housing Corporation Building, 25 Rutland Street, Auckland.

33 Kings Street, Frankton, Hamilton.

25-27 Mercer Street, Wellington.

Mulgrave Street, Wellington.

147 Hereford Street, Christchurch.

Shop 3, 279 Cuba Street (P.O. Box 1214), Palmerston North.

Cargill House, 123 Princes Street, Dunedin.

#### Other issues of the Gazette:

Commercial Edition—Published weekly on Wednesdays.

Customs Edition—Published weekly on Tuesdays.

Special Editions and Supplements—Published as and when required.

### **Advertising Rates**

The following rates for the insertion of material in the Principal Edition of the *New Zealand Gazette* apply as from 1 July 1991:

Category 1

Single column notices, e.g.: Notices under the Transport Act, Public Works Act — 55c per word.

Category 2

Notices in table form or taking up two columns across the page, e.g.: Regulation Summaries, notices under the Medicines Act - 60c per word.

The appropriate rate to be applied to an advertisement will be determined at the time of setting up the notice for publication. Customers will be invoiced in accordance with standard commercial practices. Advertising rates are not negotiable.

All rates shown are inclusive of G.S.T.

### **Parliamentary Summary**

### **Bills Assented To**

Government Bins	Assent No
31 July 1991	rissell 140
Legal Services Law Practitioners Amendment Customs Amendment Tariff Amendment Income Tax Amendment (No. 4) Stamp and Cheque Duties Amendment (No. 2)	71 72 73 74 75 76
1 August 1991	
Social Welfare (Transitional Provisions) Amendment (No. 2) Social Security Amendment (No. 2) Disabled Persons Community Welfare Amendment Rehabilitation Act Repeal Inland Revenue Department Amendment (No. 2)	77 78 79 80 81
2 August 1991	
Area Health Boards Amendment (No. 2) Social Security Amendment (No. 3)	82 83

#### **Bills Introduced**

#### **Government Bills**

(Minister/Member in Charge Shown in Parenthesis) 30 July 1991

Appropriation Bill (No. 4) (Hon. Ruth Richardson) Social Welfare Reform Bill (Hon. Jenny Shipley) Finance (Revenue) Bill (Hon. Wyatt Creech) Health Reform Bill (Hon. Simon Upton) Finance Bill (No. 2) (Hon. Maurice McTigue)

#### Referred to Select Committee

Money Bill—not referred Not referred—under urgency Not referred—under urgency Not referred—under urgency Not referred—under urgency

### **Summary of Bills Introduced**

Appropriation (No. 4)

This Bill is to issue and apply certain sums of public money for the year ending 30 June 1992, to authorise costs incurred, and to appropriate the supply granted for that year.

#### Social Welfare Reform

The significant provisions of Part I of the Bill are as follows:

- (a) An increase in the age of entitlement to domestic purposes benefits for sole parents to 18 years;
- (b) The restriction of entitlement to domestic purposes benefits to one parent in cases of split custody;
- (c) An increase in the age of entitlement to invalids' and sickness benefits to 16 years;
- (d) An inclusion of the accommodation benefit to which an applicant would otherwise be entitled in the formula for determining the period of non-entitlement of a person who has received a redundancy payment;
- (e) The introduction of provision for participation in the community task force scheme;
- (f) The reform of funeral grants;
- (g) The restriction of entitlement to benefits to one parent in cases of shared custody;
- (h) The reduction of the period during which a beneficiary may remain overseas without effecting eligibility for a benefit to 4 weeks:
- (i) The extension of the provisions relating to periods of non-entitlement for high-income earners to applicants for widows', domestic purposes, invalids', sickness, and training benefits;
- (i) The introduction of regulation-making empowering provisions relating to entitlement cards;
- (k) The reversion of the name of guaranteed retirement income to national superannuation;
- (I) The increase in the age of entitlement to guaranteed retirement income and veterans' pensions to 61 years on 1 April 1992, and the increase in the age of entitlement from 61 years to 65 years by 3-monthly increments every 6 months from 1 July 1993:
- (m) The decrease in the alternative rate of guaranteed retirement income and veterans' pensions for persons with non-qualifying spouses;
- (n) The introduction of income-testing of guaranteed retirement income;
- (o) The introduction of a provision permitting all persons who qualify for both guaranteed retirement income and a veteran's pension to transfer from guaranteed retirement income to a veteran's pension;
- (p) The introduction of income and assets testing in relation to the granting of assistance under the Disabled Persons Community Welfare Act 1975;
- (q) The repeal and revocation of the Rehabilitation Act 1941 and various related Acts, regulations, and orders.

Part II of the Bill makes amendments to the Inland Revenue Department Act 1974 to allow the Inland Revenue Department to supply to the Department of Social Welfare all or any of the following information that is held by the Inland Revenue Department in relation to a credit of tax under Part XIA of the Income Tax Act 1976 for the year ending with the 31st day of March 1991:

- (a) The names and addresses of persons in receipt of that credit of tax; and
- (b) The number of children to which that credit of tax relates; and
- (c) The amount of that credit of tax.

The purpose of the amendments is to facilitate the issuing of entitlement cards by the Department of Social Welfare.

#### Finance (Revenue)

This Bill amends the Customs Act 1966, the Tariff Act 1988, the Income Tax Act 1976 and the Stamp and Cheque Duties Amendment Act 1971 to give effect to Budget announcements in relation to:

- (a) The excise duty and other duty payable on alcohol, tobacco and motor spirits;
- (b) The application of the non-resident withholding tax provisions in respect of payments of interest by approved issuers; and
- (c) The provision of an approved issuer levy at the rate of 2 cents for each \$1 of leviable value in respect of loan transactions registered with the Commission of Inland Revenue by persons who are approved issuers under new section 311B of the Income Tax Act 1976.

#### Health Reform

This Bill amends the Area Health Boards Act 1983 and Part II of the Social Security Act 1964.

Part I which amends the Area Health Boards Act 1983:

- (a) Reconstitutes each health board as a corporation sole (consisting of a single commissioner appointed by the Minister of Health), and makes the necessary consequential amendments;
- (b) Provides that each commissioner may, with the consent of the Minister of Health, appoint a deputy; and
- (c) Imposes two qualifications on the duty of the general manager to act independently in matters relating to decisions on individual employees.

Those qualifications are:

- (i) That the State Services Commissioner may from time to time, after consultation with the Director-General of Health, promulgate conditions of employment to any general manager (not being conditions that are inconsistent with any terms and conditions of any applicable collective employment contract) and any such general manager shall ensure that the conditions of employment entered into with any person employed by the board are consistent with the conditions of employment so promulgated; and
- (ii) That the State Services Commissioner may from time to time declare that all or any part of the conditions of employment fixed under a collective employment contract for persons employed in the Health Service are to be the actual conditions of employment.
- It is expressly provided that nothing in the new subsections authorises the reduction of the individual conditions of employment of any person employed in the Health Service immediately before the commencement of the section.

Part II of the Bill amends the Social Security Act 1967, which is administered by the Department of Health, and relates to medical and hospital benefits and other related benefits.

#### Finance Bill (No. 2)

This Bill contains a number of miscellaneous provisions including amendments to the following legislation:

#### (a) Accident Compensation Act 1982

Under the present legislation, the Commissioner of Inland Revenue is required to pay the amount of any levy payable by a self-employed person when it first appears that the levy is due and payable, whether or not it has actually been paid. The new provisions provide that payments of levies by the Commissioner to the Corporation will, from 1 August 1991, occur when the levy payments are actually received by the Commissioner, and this will apply to levies from both employers and self-employed persons. Provision is made for appropriate adjustments when the Commissioner has already paid amounts in respect of any levy received after 1 August 1991, or where the amount of any levy is written off or reduced by an assessment or reassessment.

#### (b) Customs Act 1966

The amendment provides new definitions of the terms "documents" and "duty".

#### (c) Housing Corporation Act 1977

The amendments reconstitute the Housing Corporation of New Zealand, empower the Corporation to sell or dispose of its property and makes related amendments to the Housing Corporation of New Zealand.

#### (d) State-Owned Enterprises Act 1986

Major amendments relate to the Government Supply Brokerage Corporation (N.Z.) Ltd., the Forestry Corporation of New Zealand Limited, Government Computer Services Ltd., the National Provident Management Company Limited, and Timberlands West Coast Ltd.

#### (e) Adult Adoption Information Act 1985

The amendment makes it clear that fees can be prescribed to any matter under the Act and provides an interim scale of fees to cover the period until new fees regulations come into force.

#### (f) Animals Act 1967

The major amendments remove the Crown's obligation to compensate owners of cattle slaughtered because of infection with brucellosis or tuberculosis and compensation to be paid out of levies collected under the Meat Act 1981.

#### (g) Crown Minerals Act 1991

This amendment to section 108 of the Crown Minerals Act 1991 alters and reinstates, with minor amendments, subsection (1) which was omitted by Supplementary Order Paper No. 40. A consequential amendment is also made to section 387 of the Resource Management Act 1991.

#### (h) Dog Control and Hydatids Act 1982

These amendments dissolve the National Hydatids Council and provides that all the Council's rights, assets, and liabilities or debts become assets, liabilities and debts of the Minister of Agriculture.

#### (i) Education Act 1989

The Bill dissolves the Parent Advocacy Council and provides that the Council's rights, assets, and liabilities become those of the Minister of Education.

The Bill also repeals and replaces section 65 of the Education Act 1989, and makes certain other repeals. It has the effect of repealing provisions of that Act making it compulsory for teachers in kindergartens and State schools to be registered teachers, together with provisions that provide for certain exceptions. The opportunity has been taken to repeal some spent transitional provisions relating to teacher registration.

A further amendment removes a doubt as to whether or not Boards need the consent of the Minister of Education before borrowing sums denominated in New Zealand dollars. Such consent is not needed.

The Bill also amends section 67 (1) of the principal Act to take account of the fact that grants will no longer be designated as operational activities grants.

Section 79 of the principal Act (which relates to the payment of grants to Boards) is replaced with a redrafted section abolishing the distinction between operational activities grants and teacher salaries grants, and repeals sections 80 and 81 of the principal Act (which restrict, respectively, transfers between grants of those 2 types and the payment of teacher salaries from sources other than grants). Two points should be noted in relation to Boards that do not choose (under the amendments to section 91 of the principal Act intended to be effected by this Bill) to move to bulk funding of teacher salaries. First, such Boards will effectively continue to receive a grant that is the equivalent of an operational activities grant. Secondly, such Boards will continue to be restricted (by a new section 91 (9)) in their ability to pay teacher salaries from sources other than grants.

Section 91 of the Education Act 1989 is also amended to provide for the payment of teacher salaries during the transitional period before the general introduction of bulk funding of teacher salaries, and for the trialling of bulk funding.

The new section effects a number of amendments. It replaces the provisions relating to trialling of the bulk funding of teacher salaries with provisions enabling Boards to agree with the Minister of Education to move to bulk funding. It empowers the Minister to reduce in any year the grant paid to the Board of a school whose teachers continue to be paid directly by the Secretary of Education if in the year before the Board employed more teachers, or more expensive teachers, than the Secretary was required by law to pay. And it continues the restrictions (at present in section 81 of the principal Act) on the payment out of sources other than grants of the salaries of teachers at schools whose teachers continue to be paid directly by the Secretary.

#### (j) Employment Contracts Act 1991

This Part of the Bill amends a transitional provision relating to redundancy provisions which were in force at the commencement of the Act.

#### (k) Gaming and Lotteries Act 1977

The Bill amends the definitions of "game of chance", "gaming machine", "illegal prize competition", and "illegal game of chance". Other changes include allowing the Minister of Internal Affairs to issue or amend licences to conduct games of chance (renewable annually), to seek and consider the advice from such sources (including the Police) as the Minister considers relevant, a provision authorising all licences in force before the Bill to have been issued for a period expiring on 30 September 1991, and provision for the making of regulations in relation to the issue, amendment and renewal of such licences.

#### (1) Gaming Duties Act 1971

This Part of the Bill imposes a new duty for gaming machines.

#### (m) Income Tax Act 1976

This Part of the Bill amends the Income Tax Act 1976 to give effect to the announcements in the Budget relating to—

- (i) The abolition of the guaranteed retirement income earner surcharge with effect from 1 April 1992, and the change from guaranteed retirement income to national superannuation;
- (ii) The removal of the tax exemption for inter-corporate dividends, with general effect from 1 April 1992 and immediate effect from 8 p.m. on 30 July 1991 in relation to certain categories of fixed-rate shares;
- (iii) Changes in relation to the continuity and commonality of shareholding provisions that apply to the carrying forward of losses by companies and the offsetting of losses within groups of companies, insofar as those changes affect the 1991– 92 income year; and
- (iv) The tightening up of certain anti-avoidance provisions relating to dealings in shares and to the imputation provisions of the Income Tax Act 1976.

#### (n) Land Transfer Act 1952

This amendment, which comes into force on 1 September 1991, has 2 effects. First, it replaces section 235 of the Land Transfer Act 1952 with a redrafted section making it clear that fees can be prescribed in relation to any matter under that Act, and any act done by a Registrar as the Registrar-General under any other enactment; and that in the prescription of the amount of any fee regard can be had to the costs and expenses incurred by the Department of Survey and Land Information in providing survey maintenance and control, and maintaining cadastral records, for the purposes of that Act.

Secondly, it deletes from other provisions of that Act specific references to the prescription and payment of fees.

#### (o) Maori Education Foundation Act 1961

This Part of the Bill repeals a spent permanent appropriation in section 20 of the Maori Education Foundation Act 1961.

#### (p) Meat Act 1981

The amendments prevent the Minister of Agriculture approving schemes to assist disease eradication in cattle and deer except on the recommendation of the Animal Health Board. In addition, the fixing of fees and levies cannot be made unless the Minister of Agriculture consults with the Animal Health Board and Federated Farmers.

#### (q) National Library Act 1965

These amendments in this Part align the provisions of the National Library Act with the provisions of the Public Finance Act 1989.

#### (r) Public Finance Act 1989

This Part of the Bill, modifies the application of Public Finance Act 1989 to:

- (i) The New Zealand Security Intelligence Service; and
- (ii) The Government Communications Security Bureau.

#### (s) Public Works Act 1981

The amendments in this Part require money from the sale of land be paid into a Crown Bank Account or Departmental Bank Account in accordance with the Public Finance Act 1989, rather than into the Public Account as presently provided.

#### (t) Transit New Zealand Act 1989

The major changes include changes to the amount of excise duty payable on motor spirits designated for roading purposes, and a reduction for a limited period of the amount of excise duty and goods and services tax payable on motor spirits that is refundable.

#### (u) Valuers Act 1948

The amendments in this Part provide for:

- (i) The payment of an annual registration fee by registered valuers;
- (ii) The removal of a registered valuer from the Registrar of Valuers if the annual registration fee is not paid;
- (iii) The amount of the fee to be set by regulation;
- (iv) The Valuers Registration Board to reimburse the Valuation Department and other Government agencies for costs incurred by them in the administration of the Valuers Act 1948.

#### (v) Veterinary Surgeons Act 1956

The amendments in this Part include provision for:

- (i) The Director-General of Agriculture and Fisheries to charge the Veterinary Surgeons Board for services provided by the employees of the Ministry;
- (ii) The Veterinary Surgeons Board to pay the remuneration and travelling allowances of its members out of its own funds;
- (iii) Fees for application for registration, for certificates of registration, for monetary penalties imposed in disciplinary cases (and for any additional costs incurred) and for examination fees be payable to the Board rather than to the Crown; and
- (iv) Widening the categories of payment that may be made into and of payments that made be made from the account formerly known as the Veterinary Surgeons Special Fund.

#### (w) New Zealand Planning Act 1982

This Part of the Bill dissolves the New Zealand Planning Council and repeals the New Zealand Planning Act 1982.

ps7944

### **Bills Assented To**

#### **Government Bills** Assent No. 8 August 1991 Customs Amendment (No. 2) 84 Income Tax Amendment (No. 5) 85 Transit New Zealand Amendment (No. 2) 86 87 Public Works Amendment Valuers Amendment 88 Gaming and Lotteries Amendment 89 90 Education Amendment (No. 2) 91 Housing Corporation Amendment 92 Accident Compensation Amendment (No. 2) 93 Finance (No. 2) 94 95 Adult Adoption Information Amendment Animals Amendment 96 Meat Amendment 97 New Zealand Planning Council Dissolution **Employment Contracts Amendment**

### **Bills Introduced**

#### **Government Bills**

(Minister/Member in Charge Shown in Parenthesis) 5 August 1991

Privacy of Information (Hon. D. A. M. Graham) Income Tax Amendment (No. 6) (Hon. Wyatt Creech) Referred to Select Committee

Justice and Law Reform Finance and Expenditure

(Minister/Member in Charge Shown in Parenthesis) Finance (No. 4) (Hon. Maurice McTigue) Referred to Select Committee
Government Administration

### **Summary of Bills Introduced**

Privacy of Information

This Bill sets down a number of information privacy principles. These principles relate to the collection, use, and disclosure of, and access to, personal information (i.e., information about an identifiable individual) that is held by public or private sector agencies. In general, these principles do not create legal rights enforceable in a Court.

The Bill provides for the appointment of a Privacy Commissioner. The Commissioner, in conjunction with the Equal Opportunities Tribunal (re-named the Human Rights Tribunal), provides an enforcement mechanism for the information privacy principles.

The Privacy Commissioner would also have a more general role in relation to the protection of individual privacy. In this regard, the Commissioner takes over the general watchdog role that the Human Rights Commission currently performs under Part V of the Human Rights Commission Act 1977. The Privacy Commissioner's role includes the promotion of the information privacy principles, inquiring into enactments, laws, practices, and procedures that may infringe individual privacy, the monitoring of technological developments as they relate to individual privacy, and providing assistance and advice to agencies in relation to their obligations under the Bill.

Parts IV and V of the Bill relate to the information privacy principles that confer entitlements to access to personal information. They set out the grounds on which such access may be refused, and prescribe the procedures that agencies must follow in dealing with requests for such access. The provisions are modelled on Parts II and IV of the Official Information Act 1982.

Part VI provides for exemptions from the information privacy principles.

Part VII sets out the procedures for dealing with possible cases of interference with the privacy of individuals. The initial investigation is carried out by the Privacy Commissioner, either on a complaint or on his or her own initiative. After investigation, the Commissioner is to attempt to secure a settlement of the matter and obtain an assurance against repetition of the interference. If that fails, proceedings may be taken before the Human Rights Tribunal against the person alleged to have interfered with an individual's privacy. The Tribunal will have power to grant a number of remedies, including an award of damages.

Part VIII sets out the powers of the Commissioner in carrying out an investigation under Part VII of the Bill.

Part IX contains provisions relating to information matching. Information matching consists of the comparison of one set of personal information with another set of personal information.

The provisions of *Part IX* provide for the approval, by the Privacy Commissioner, of information matching programmes that would otherwise be a breach of the information privacy principles, or are prohibited by statute. *Part IX* also lays down certain requirements that will apply to information matching programmes (whether approved by the Privacy Commissioner or authorised by statute), except for programmes that are already authorised by statute.

Part X relates to unique identifiers. These are identification numbers or other identifiers that are assigned to an individual by an agency in order to identify the individual in dealings with the agency.

Part X provides for the making of regulations regulating the creation and use of unique identifiers. Such regulations may prescribe the procedures to be used by agencies in assigning unique identifiers to individuals, and may impose restrictions on their use.

Part XI makes special provision for certain law enforcement information. That Part incorporates, in a modified form, the Schedule to the Wanganui Computer Centre Act 1976 (which is repealed by the Bill). The purpose of Part XI is to authorise access by certain Government departments and local authorities to law enforcement information stored by certain Government departments on the Wanganui Computer.

Part XII contains various miscellaneous provisions relating to the operation of Parts II to VIII of the Bill. It also provides for the repeal of certain enactments, including the Wanganui Computer Centre Act 1976, and various provisions of the Area Health Boards Act 1983, the Health Act 1956, and the Hospitals Act 1957.

Part XIII contains a number of amendments to various enactments. These amendments fall into two categories—

- (a) Amendments that are consequential on the provisions of the Bill:
- (b) Amendments that authorise the exchange of information between Government agencies for the purposes of preventing and detecting fraud and abuse of the social welfare system.

Income Tax Amendment (No. 6)

This Bill amends the Income Tax Act 1976 in a number of respects.

The main areas of amendment are as follows:

- (a) A number of changes are made in the definition of the term "dividends" and related definitions in sections 3 to 4B of the Act, with numerous consequential and related amendments throughout the Bill. These changes largely follow the recommendations of the Valabh Committee.
- (b) The continuity and commonality of shareholding provisions in the Act (sections 188 and 191) that relate to the ability of companies to carry forward losses and set off losses between group members are repealed and replaced with new provisions that (i) are based on a standard continuity percentage of 66 percent, and (ii) now require 66 percent commonality of interest throughout an income year before the company loss offset provisions can be utilised.

The standard continuity percentage of 66 percent is also applied to the shareholding continuity requirements relating to imputation credit accounts and other accounts in sections 394E (1) (g), 394zw (1) (f), and 394zzp (3) (d) of the Act.

(c) New Part XIIAA introduces the qualifying company regime recommended by the Valabh Committee in relation to closely-held companies. Under the regime, companies with (generally) 5 shareholders or less who meet the election and other

- requirements for becoming a qualifying company, will, together with their shareholders, be subject to a tax regime more similar to that applying to partnerships.
- (d) The new section 276 relates to companies left with insufficient assets to meet their tax liabilities, and makes directors, controlling shareholders, and in some circumstances other shareholders of such companies liable to pay the company's unsatisfied tax debts in certain circumstances where it appears that an arrangement was entered into for the purpose of leaving the company unable to satisfy its tax liabilities.
- (e) Clause 8 proposes a 2-year extension of the now limited inter-corporate dividend tax exemption in section 63 of the Act in the case of dividends payable on certain kinds of redeemable fixed-rate shares acquired before 8 p.m. on 30 July 1991.
- (f) Clause 2 proposes new definitions of the terms "local authority" and "local authority trading enterprise", and clause 36 amends the definition of the latter term in the Local Government Act 1974.

Finance (No. 4)

This Bill dissolves the New Zealand Export-Import Corporation and repeals the New Zealand Export-Import Corporation Act 1974. The Bill also proposes to amend the following Acts:

- (a) The Official Information Act 1982 and the Ombudsmen Act 1975 by defining related companies of State enterprises and making them subject to those Acts;
- (b) The Reserves Act 1977 by changing the financial year of the Reserves Board and other Boards appointed under the Act;
- (c) The State-Owned Enterprises Act 1986 by:
  - (i) Empowering the Governor-General by Order in Council to add to the list of State enterprises in the First and Second Schedules to the principal Act;
  - (ii) Altering the content required and delivery times for draft statements of corporate intent of State enterprises, and delivery times of the completed statements to the shareholding Ministers;
    - (iii) Simplifying the procedure for the transfer of land to State enterprises;
  - (iv) Making consequential amendments to the Treaty of Waitangi Act 1975 and various sections of the State-Owned Enterprises Act 1986; and
  - (v) Providing that nothing in that Act prevents the exercise, in relation to a State enterprise named in the Second Schedule to the Act, of a power conferred by section 23 or section 24 or section 28 of the Act.

The Bill further seeks to amend:

- (d) The Transport Services Licensing Act 1989 by abolishing the Passenger Complaints Board; and
- (e) The Veterinary Surgeons Act 1956 by removing the Council of the New Zealand Veterinary Surgeons Association from involvement in the preliminary investigation of formal complaints against veterinary surgeons.

ps8088

### **Government Notices**

### Commerce

#### **Electricity Act 1968**

### Notice of Issue of Draft Electrical Code of Practice for Hazardous Area Installations.

Public comment is sought on draft document ECP:24 which it is proposed to issue as an electrical code of practice pursuant to the Electricity Act 1968.

The draft document outlines requirements for hazardous area installations.

Copies of the draft may be obtained by writing to "ECP:24 for comment", Chief Electrical Inspectors Office, Energy and Resources Division, Ministry of Commerce, P.O. Box 2337, Wellington.

Closing date for receipt of comments is on the 11th day of October 1991.

Dated this 1st day of August 1991.

P. J. MORFEE, Chief Electrical Inspector. go8094

#### Conservation

#### **Harbours Act 1950**

### The New Plymouth District Council Waitara Foreshore and River Control Order 1991

CATHERINE A. TIZARD, Governor-General ORDER IN COUNCIL

At Wellington this 5th day of August 1991

Present:

HER EXCELLENCY THE GOVERNOR-GENERAL IN COUNCIL

Pursuant to sections 8A and 165 of the Harbours Act 1950, Her Excellency the Governor-General, on the recommendation of the Minister of Conservation, and acting by and with the advice and consent of the Executive Council, hereby makes the following order.

#### Order

- 1. Title and commencement—(1) This order may be cited as the Waitara Foreshore and River Control Order 1991.
- (2) This order shall come into force on the 28th day after the date of its publication in the *Gazette*.
- 2. Interpretation—In this order—

- "The Act" means the Harbours Act 1950:
- "The Council" means the New Plymouth District Council:
- "Foreshore" means such parts of the bed, shore or banks of a tidal water as are covered and uncovered by the flow and ebb of the tide at ordinary spring tides.
- **3. Grant of Control**—There is hereby granted to the council for a period of 21 years from the commencement of this order, the control of the water and foreshores described in the Schedules to this order.
- **4. Conditions of Grant**—The grant of control conferred by clause 3 of this order is subject to the following conditions—
- (a) Suitably worded signs shall be erected at main public access ways to the area described in the Schedule to this order indicating that control of it has been granted to the council pursuant to section 165 of the Act.
- (b) The rights and privileges conferred by this Order in Council shall not apply to those portions of the foreshore required for the securing of shore ends of any telephone cables that are at present or may be at any time laid down within the foreshore described in the First Schedule to this order.
- (c) The council may enclose any part or parts of the foreshore described in the First Schedule of this order for the purpose of holding athletic sports or games and may by bylaw fix a charge for admission to such enclosed part or parts.
- (d) Subject to the provisions of section 8A of the Act, the council for and within the area of waters described in the Second Schedule to this order
  - (a) may by bylaw do anything which a harbour board may do under section 232 of the Act, and
  - (b) may appoint harbourmasters and/or officers and define or limit their powers or duties.

#### First Schedule

All those parts of the foreshore of the Waitara River bounded to the east by a line being the production of the western edge of Hutchins Street and to the west by the production of a line from the western boundaries of Lots 2 and 3, Deposited Plan 14577 in the Township of Waitara as shown in plan No. DOC(CM) WA00050, held in the office of the Wanganui Conservancy.

#### Second Schedule

All that area of water in the Waitara River having as its inner limits a line across the river 80 metres south of the southern edge of Brookes Terrace, opposite Brown Street in the township of Waitara and having as it's outer limites a line drawn between the seaward ends of the training walls as shown in plan No. DOC(CM) WA00050, held in the office of the Wanganui Conservancy.

MARIE SHROFF, Clerk of the Executive Council. go8118

### The New Plymouth District Council New Plymouth Foreshore Control Order 1991

CATHERINE A. TIZARD, Governor-General ORDER IN COUNCIL

At Wellington this 5th day of August 1991

Present:

HER EXCELLENCY THE GOVERNOR-GENERAL IN COUNCIL

Pursuant to section 165 of the Harbours Act 1950, Her Excellency the Governor-General, on the recommendation of the Minister of Conservation, and acting by and with the advice and consent of the Executive Council, hereby makes the following order.

#### Order

**1. Title and commencement**—(1) This order may be cited as the New Plymouth Foreshore Control Order 1991.

- (2) This order shall come into force on the 28th day after the date of its publication in the *Gazette*.
- 2. Interpretation—In this order—
  - "The Act" means the Harbours Act 1950:
  - "The Council" means the New Plymouth District Council:
  - "Foreshore" means such parts of the bed, shore or banks of a tidal water as are covered and uncovered by the flow and ebb of the tide at ordinary spring tides.
- **3. Grant of Control**—There is hereby granted to the council for a period of 21 years from the commencement of this order, the control of the water and foreshores described in the Schedules to this order.
- **4. Conditions of Grant**—The grant of control conferred by clause 3 of this order is subject to the following conditions—
- (a) Suitably worded signs shall be erected at main public access ways to the area described in the Schedule to this order indicating that control of it has been granted to the council pursuant to section 165 of the Act.
- (b) The rights and privileges conferred by this Order in Council shall not apply to those portions of the foreshore required for the securing of shore ends of any telephone cables that are at present or may be at any time laid down within the foreshore described in the First Schedule to this order.
- (c) The council may enclose any part or parts of the foreshore described in the First Schedule of this order for the purpose of holding athletic sports or games and may by bylaw fix a charge for admission to such enclosed part or parts.

#### Schedule

All that area in the Taranaki Land District comprising that part of the foreshore (being also part New Plymouth Harbour Reserve A comprised in deeds index 4, folio 585) bounded to the east by a line being the production of the south-western boundary of Section 200, Hua District, and to the west by a line being the production of the south-western side of Belt Road in the City of New Plymouth; part of the said area being also part of the New Plymouth Harbour Roadstead, containing in certificate of title, Volume 102, folio 148, Taranaki Registry as shown in plan No. DOC(CM) WA00051.

MARIE SHROFF, Clerk of the Executive Council.

### **Education**

#### **Education Act 1989**

#### Te Kura Kaupapa Maori O Waipareira Amendment Notice 1991

Pursuant to section 155 (8) of the Education Act 1989 (as inserted by section 14 of the Education Amendment Act 1989), the Minister of Education hereby gives the following notice:

#### Notice

1. The Te Kura Kaupapa Maori O Waipareira 1990 Notice is hereby amended by the insertion of the following new clause:

#### 5A Constitution of the Board of Trustees

- (a) 5 parent representatives who shall be people who are parents of students enrolled full-time at the school when the roll for the election (or, where there is a supplementary roll, the supplementary roll) closes; elected from and among themselves; and
- (b) the principal; and
- (c) a staff representative who shall be a person (other than the principal) who, on the day on which the roll for the election (or, where there is a supplementary roll, the supplementary roll) closes is a member of the

Board staff elected by people (other than the principal) who are members of the Board staff on that day; and

- (d) not more than 4 persons coopted by the Board after consultation with the kura whanau.
- 2. Clause 5 of the notice entitled Te Kura Kaupapa Maori O Waipareira 1990 published in the Gazette on 25 January 1990, No. 9 at page 255 is hereby revoked.
- 3. This notice shall come into force on the day after the date of its publication in the Gazette.

Dated this 23rd day of July 1991.

C. GIBSON, Group Manager Operations. go8113

#### Early Childhood Centre Exempt From Licensing

Pursuant to section 316 (1) (b) of the Education Act 1989 (as inserted the Education Amendment Act 1990), I hereby give the following notice:

The following early childhood centres are exempt from the requirements to be licensed as an early childhood centre:

YMCA Mobile Pre-School Unit Clevedon Mobile Pre-School Unit Franklin Mobile Pre-School Unit Thames Mobile Pre-School Unit Lakeland Mobile Pre-School Unit Hunterville Mobile Pre-School Unit Marlborough Mobile Pre-School Unit Waimate Mobile Pre-School Unit Southland Mobile Pre-School Unit

This notice shall come into force on the day after the date of its notification in the Gazette.

JOHN LUXTON, for Minister of Education.

go8117

#### Health

#### **Medicines Act 1981**

#### Consent to the Distribution of Changed Medicines

Pursuant to section 24 (5) of the Medicines Act 1981, the Minister of Health hereby consents to the distribution in New Zealand of the changed medicines set out in the Schedule hereto:

#### Schedule

Name and Strength Ciprofloxacin 100 mg/50 ml, 200 mg/100 ml Phase 1 tablets (white) estradiol valerate 2 mg Phase 2 tablets (pink) estradiol valerate 2 mg/

levonorgestrel 0.075 mg

Form

**Tablets** 

Solution for infusion (vials)

Bayer AG, Leverkusen, Germany

Name and Address of Manufacturer

Schering AG, Berlin, Germany

Proprietary Name

(if any) Ciproxin IV 100 Ciproxin IV 200 Nuvelle

Dated this 7th day of August 1991.

MAURICE WILLIAMSON, Minister of Health.

#### Revocation of Consent to the Distribution of a New Medicine

Pursuant to section 35 of the Medicines Act 1981, the Minister of Health hereby revokes consent to the distribution in New Zealand of the new medicine set out in the Schedule hereto:

#### Schedule

Name and Strength Spironolactone 25 mg,

Form

Name and Address of Manufacturer

Proprietary Name (if any)

100 mg

**Tablet** 

Tablet Manufacturers Ltd., Takapuna, Auckland

Dated this 7th day of August 1991.

MAURICE WILLIAMSON, Minister of Health. qo8091

#### Consent to the Distribution of New Medicines

Pursuant to section 20 of the Medicines Act 1981, the Minister of Health hereby consents to the distribution in New Zealand of the new medicines set out in the Schedule hereto:

#### Schedule

Name and Strength	Form	Name and Address of Manufacturer	Proprietary Name (if any)
Soya oil 10% w/v medium chain triglycerides 10% w/v egg lecithin 1.20% w/v glycerol 2.50% w/v	Fat emulsion for intravenous infusion	B Braun Melsungen AG, Melsungen, Germany	Lipofundin MCT/LCT 20%
Soya oil 5% w/v medium chain triglycerides 5% w/v egg lecithin 1.20% w/v glycerol 2.50% w/v	Fat emulsion for intravenous infusion	B Braun Meisungen AG, Melsungen, Germany	Lipofundin MCT/LCT 10%
Etoposide 100 mg/5 ml	Solution for injection (ampoule & vial)	David Bull Laboratories Pty Ltd., Mulgrave, Victoria, Australia	
Folic acid 15 mg/1 ml	Solution for injection (ampoule)	Abbott Australasia Pty Ltd., Kurnell, New South Wales, Australia	
Albendazole 400 mg	Tablet	Smith Kline & French Laboratories Ltd., Mundells, Welwyn Garden City, England, or Laboratoires Smith Kline & French SA, Pessac, France	Eskazole
Fosinopril 5 mg, 10 mg, 20 mg	Tablet	E R Squibb & Sons Pty Ltd., Noble Park, Victoria, Australia, or E R Squibb & Sons Ltd., Moreton, Wirral, Merseyside, England	Monopril
Calcium alginate fibre 80% Sodium alginate fibre 20%	Sterile wound dressing	Britcair Ltd., Rhymney, Gwent, Wales	Kaltoclude
Nicotine 0.5 mg, 1 mg, 1.5 mg	Transdermal patch	Alza Corporation, Palo Alto, California, United States of America, or Alza Corporation, Vacaville, California, United States of America	Nicoderm TTS
D : 1:1: 5:1 1 (A	1001		

Dated this 7th day of August 1991.

MAURICE WILLIAMSON, Minister of Health. go8089

### **Internal Affairs**

#### Queen Elizabeth the Second Arts **Council of New Zealand Act 1974**

#### **Establishment of Whangaroa Community Arts** Council

Pursuant to section 32 (1) of the Queen Elizabeth the Second Arts Council of New Zealand Act 1974, on the recommendation of the Northern Regional Arts Council, the Queen Elizabeth II Arts Council of New Zealand hereby

Title

establishes the Whangaroa Community Arts Council for the following duly defined area:

the Whangaroa Ward.

Dated at Wellington this 20th day of June 1991.

The Seal of the Queen Elizabeth II Arts Council of New Zealand affixed in the presence of:

- J. PATTRICK, Chairman.
- C. WHITING, Deputy Chairman.
- M. M. JENSEN, Witness. go8093

#### **Justice**

#### **Indecent Publications Act 1963**

### **Indecent Publications Tribunal Decision**

On 14 February 1991, the Indecent Publications Tribunal considered the following publication:

Decision No.

> 27/91 31/7/91 Urban Aboriginals: A Celebration of Leather Sexuality

Date

Gay Sunshine Press

Publisher

Lawrence Publishing

*Importer* 

Applicant

Decision Comptroller of Not indecent

Customs

Copies of these decisions containing reasons may be purchased from Tribunals Division, P.O. Box 10-044, Wellington, at a cost of \$6.00 each or by annual subscription of \$90.00.

N. KERSHAW, for Secretary, Indecent Publications Tribunal. 908095

Decision No. 10/91

Reference No.: IND 26/90

#### Before the Indecent Publications Tribunal

In the matter of the Indecent Publications Act 1963, and in the matter of an application by the Comptroller of Customs for a decision in respect of the two publications contained in application No. 26/90 both of which have the title: *The Lingerie Model.* Publisher: American Art Enterprises:

Chairperson: P. J. Cartwright.

 $\label{eq:Members: R. E. Barrington, S. C. Middleton and W. K. Hastings.$ 

Hearing at Auckland on the 14th day of February 1991.

Appearances: B. N. Cheesman on behalf of the importer. M. J. Wotherspoon on behalf of Comptroller of Customs.

#### Corrigendum

This corrigendum is issued to clarify the classification of the two publications both titled *The Lingerie Model* in decision No. 10/90 issued on 31 May 1991.

The Lingerie Model by Alice Adrian is classified as unconditionally indecent. The Lingerie Model by Carl Benedict is classified as indecent in the hands of persons under the age of 18 years.

Dated at Wellington this 8th day of August 1991.

P. J. CARTWRIGHT, Chairperson.

Indecent Publications Tribunal. go8109

#### **Judicature Act 1908**

#### Acting Judge of High Court Appointed

Pursuant to section 11A of the Judicature Act 1908, Her Excellency the Governor-General, in the name and on behalf of Her Majesty the Queen, has been pleased to appoint

The Honourable Sir Graham Davies Speight

to be an acting Judge of the High Court for a term of 1 year, the Commission to take effect on the 1st day of September 1991.

Dated at Wellington this 5th day of August 1991.

PAUL EAST, Attorney-General. go8092

#### **Justices of the Peace Act 1957**

#### Notification of Change of Name

It is noted for information that

Joy Dickinson, Justice of the Peace of Christchurch

has changed her name by deed poll to

Joy Ashenden.

Dated at Wellington this 6th day of August 1991.

D. OUGHTON, Secretary for Justice.

(Adm. 3/17/9) go8121

#### Marriage Act 1955

#### Marriage Celebrants for 1991, Notice No. 21

Pursuant to the provisions of section 11 of the Marriage Act 1955, the following names of marriage celebrants within the meaning of the said Act are published for general information:

Warner, Verna Lenice, 18 O'Reilly Crescent, Taumarunui.

Wilkie, Maureen Helen, Te Kumi Road, Te Kuiti.

Dated at Lower Hutt this 2nd day of August 1991.

B. E. CLARKE, Registrar-General. go8101

#### Marriage Celebrants for 1991, Notice No. 22

Pursuant to the provisions of section 13 of the Marriage Act 1955, it is hereby notified that the following names have been removed from the list of marriage celebrants under section 10 of the Act:

Au, Ioapo Paese, The Church of Jesus Christ of Latter-Day Saints.

Harvey, Kevin Desmond, Jehovah's Witness.

Higgins, Raymond Douglas, The Church of Jesus Christ of Latter-Day Saints.

Ioder, Glenn Karl, Methodist.

Kendall, Robert Harry, Jehovah's Witness.

Kohe, Rangiriri, The Church of Jesus Christ of Latter-Day Saints.

Rangitutia, Pukerau, Methodist.

Dated at Lower Hutt this 5th day of August 1991.

B. E. CLARKE, Registrar-General. go8102

#### Marriage Celebrants for 1991, Notice No. 23

Pursuant to the provisions of section 10 of the Marriage Act 1955, the following names of marriage celebrants within the meaning of the said Act are published for general information:

Andrews, Patrick Hikamori, Faith Fellowship.

Braun, Craig William, Baptist.

Deane, Kenneth Edward, The Salvation Army.

Dollin, Peter, The Salvation Army.

Faamausili, Pouniu, Congregational Christian Church in Samoa.

Felt, Leon, The Church of Jesus Christ of Latter-Day Saints.

Fruean, Luisa Ena, Presbyterian.

Fruean, Malutafa Fa'alili, Presbyterian.

Gago, Gago S., Assembly of God Church of Samoa.

Gibson, Ruth, Baptist.

Greening, Te Waata Reid, The Church of Jesus Christ of Latter-Day Saints.

Havilitama, Steve Katene, Presbyterian.

Hoddinott, Rosalie June, Methodist.

Kelly, Graham, Christian Renewal Fellowship.

Kingsley, David Calvin, Presbyterian.

McGeek, Neville, Presbyterian.

Major, Stephen Lewis, Baptist.

Mau'u, Filimone, Assembly of God in New Zealand.

Meagher, Desmond, Roman Catholic.

Moss, Thomas Murray, Jehovah's Witness.

Murray, Paul Jeffrey, Jehovah's Witness.

Norrish, Paul Wayne, Baptist.

Osborne, Paul, The Church of Jesus Christ of Latter-Day

Parkes, Judith Rae, The Methodist Church of New Zealand.

Pearce, Te Awa Will, The Church of Jesus Christ of Latter-Day Saints.

Prins, Dirk, The Auckland Central Christadelphians.

Rowell, Graham Thomas, Associated Churches of Christ in New Zealand.

Scarlett, Allan Trevor, Assembly of God in New Zealand.

Smith, Edward, Jehovah's Witness.

Spinney, Bryan Frank, Methodist.

Stevenson, Mervyn Hugh, Assembly of God in New Zealand.

Taula, Iosefa, Assembly of God in New Zealand.

Valele, Simo, Brethren.

Walker, Peter Robert, Church of Christ.

Dated at Lower Hutt this 5th day of August 1991.

B. E. CLARKE, Registrar-General.

go8103

4

### **Authorities and Other Agencies of State**

# **Broadcasting Standards Authority**

#### **Broadcasting Act 1989**

### Broadcasting Standards Authority—Decision No. 34/91

Pursuant to section 15 of the Broadcasting Act 1989, notice is hereby given that the Broadcasting Standards Authority has made the following Decision on a complaint referred to it for investigation and review.

In Decision 34/91, the Authority declined to uphold a complaint by Ms Jean Creighton of Auckland that the

broadcast by Television New Zealand Limited of three advertisements for Galaxy Cheese on a number of occasions in September 1990 breached the responsibility placed on broadcasters not to broadcast advertisements which encourage belief in inaccurate or outdated stereotypes but upheld the complaint that one of the advertisements breached the responsibility placed on broadcasters not to broadcast advertisements which use women's sexual appeal simply to draw attention to an unrelated product.

Copies of Decisions may be purchased from the Broadcasting Standards Authority, P.O. Box 9213, Wellington this 9th day of August 1991.

G. POWELL, Executive Officer.

au8122

### **Land Notices**

### Conservation

#### **Conservation Act 1987**

#### Declaring Land to be Held for Conservation Purposes Together With a Declaration That the Area Become a Recreation Reserve

Pursuant to section 7 (1) of the Conservation Act 1987, the Minister of Conservation and the Minister of Lands being the Minister responsible for the Department of State that has control of the land described in the Schedule hereto, jointly declare that the land is held for conservation purposes and pursuant to section 8 (1A) of the Act, the Minister of Conservation further declares that the conservation area shall hereafter be set apart as a reserve, subject to the Reserves Act 1977 and classified as a recreation reserve, subject to the provisions of the latter Act.

#### Schedule

#### Otago Land District—Queenstown-Lakes District

9.6 hectares, more or less, being Section 1, S.O. Plan 23397, situated in Block XIX, Shotover Survey District, subject to a

permit issued under section 66A, Land Act 1948 and dated the 8th day of February 1989.

Dated at Wellington this 22nd day of July 1991.

ROB STOREY, Minister of Lands.

DENIS MARSHALL, Minister of Conservation.

(DOC C.O. CMO 9/20)

ln8111

#### Declaration That a Marginal Strip is Not Required

Pursuant to section 24B of the Conservation Act 1987, the Minister of Conservation, hereby declares that the marginal strip requirements set out in section 24 of the Act shall not apply in the case of the disposition of the land in the Schedule hereto.

#### Schedule

That part of the former Lynmore Farm Settlement, situated along Lagoon Creek in part Section 10 of Run 301C, Block II, Mararoa Survey District.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation. In 8080

2

#### Declaring Conservation Land to be a Reserve

Pursuant to section 8 (1A) of the Conservation Act 1987, the Minister of Conservation hereby declares the conservation area, described in the Schedule hereto, to be set apart as a reserve, subject to the Reserves Act 1977 and classified as an historic reserve, subject to the provisions of the latter Act.

#### Schedule

#### Otago Land District—Clutha District

217.11 hectares, more or less, being Section 1, S.O. Plan 22545, situated in Block XI, Tuapeka East Survey District. New Zealand Gazette, 1990, page 3169.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

(DOC H.O. REH 0189, C.O. HIS 15)

2/1

#### Declaring Conservation Land to be a Reserve

Pursuant to section 8 (1A) of the Conservation Act 1987, the Minister of Conservation hereby declares that the conservation area described in the Schedule hereto shall be set aside as reserve, subject to the Reserves Act 1977 and classified as a local purpose (community facility) reserve, and further pursuant to section 26 (1) of the Reserves Act 1977, declares that the said area is hereby vested in The Marlborough District Council in trust for that purpose.

#### Schedule

#### Marlborough Land District—Marlborough District Council

6138 square metres, more or less, being Section 233, Town of Havelock, situated in Block XII, Wakamarina Survey District.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

(H.O. ORE 0099, C.O. RES: 612) In8082

#### Declaring Land to be Held for Conservation Purposes Together With a Declaration That the Area Become a Local Purpose (Cemetery) Reserve

Pursuant to section 7 (1) of the Conservation Act 1987, the Minister of Conservation and the Minister of Lands, being the Minister responsible for the Department of State that has control of the land described in the Schedule hereto, jointly declare that the land is held for conservation purposes and pursuant to section 8 (1A) of the Act, the Minister of Conservation further declares that the conservation area shall hereafter be set apart as a reserve subject to the Reserves Act 1977 and classified as a local purpose (cemetery) reserve subject to the provisions of the latter Act, and further, vests the said reserve in The Rodney District Council in trust for that purpose.

#### Schedule

#### North Auckland Land District—Rodney District

124 square metres, more or less, being Allotment 277, Tauhoa Parish, situated in Block V, Mahurangi Survey District. All Proclamation 413180.1 (New Zealand Gazette, 1975, page 2137), S.O. Plan 46846.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

W. ROB STOREY, Minister of Lands.

(Files R.C. DOC 329 H.O. ORE 0025) In8083

#### Declaring Land to be Held for Conservation Purposes Together With a Declaration That the Area Become a Recreation Reserve

Pursuant to section 7 (1) of the Conservation Act 1987, the Minister of Conservation and the Minister of Lands, being the Minister responsible for the Department of State that has control of the land described in the Schedule hereto, jointly declare that the land is held for conservation purposes and pursuant to section 8 (1A) of the Act, the Minister of Conservation further declares that the conservation area shall hereafter be set apart as a reserve subject to the Reserves Act 1977 and classified as a recreation reserve subject to the provisions of the latter Act, and further, vests the said reserve in The Manukau City Council in trust for that purpose.

#### Schedule

#### North Auckland Land District—Manukau City

700 square metres, more or less, being Section 31, Block V, Otahuhu Survey District, S.O. Plan 54520.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

W. ROB STOREY, Minister of Lands.

(Files R.O. LRC. 760, H.O. RRC 0097)

#### Declaring of Land as Conservation Area

Pursuant to section 7 (1) of the Conservation Act 1987, the Minister of Conservation and the Minister of Lands, being the Minister responsible for the Department of State that has control of the land jointly, declare that the land described in the Schedule hereto is held for conservation purposes and it shall hereafter be so held.

#### Schedule

#### Wellington Land District—Ruapehu District

5666 square metres, more or less, being Lot 7, D.P. 3791, situated in Block VI, Makotuku Survey District. Balance Gazette notice 443524.1.

Dated at Wellington this 31st day of July 1991.

DENIS MARSHALL, Minister of Conservation.

W. ROB STOREY, Minister of Lands.

(DOC C.O. GO2/115)

#### Declaring Land to be Held for Conservation **Purposes**

Pursuant to section 7 (1) of the Conservation Act 1987, the Minister of Conservation and the Minister of Lands, being the Minister deemed responsible for the Department of State that has control of the land described in the Schedule hereto, jointly declare that the land is held for conservation purposes and it shall hereafter be so held.

#### Schedule

#### Nelson Land District—Buller District

8096 square metres, more or less, being Lots 8, 9, 10 and 11, D.P. 38 and Lots 61, 62, 63 and 64, D.P. 40, situated in Block XIV, Reefton Survey District. All New Zealand Gazette, 1955, page 1590 (Gazette notice 2075).

Dated at Wellington this 22nd day of July 1991.

W. ROB STOREY, Minister of Lands.

DENIS MARSHALL, Minister of Conservation.

(File Cons. C.O. 1/25/1/312) In8086

2

2

2

#### Declaring Land to be Held for Conservation Purposes Together With a Declaration That the Same Land Become Scenic Reserve

Pursuant to section 7 (1) of the Conservation Act 1987, the Minister of Conservation and the Minister of Lands, being the Minister deemed responsible for the Department of State that has control of the land described in the Schedule hereto, jointly declare that the land is held for conservation purposes and pursuant to section 8 (1A) of the Act, the Minister of Conservation further declares that the conservation area shall hereafter be held as a reserve for scenic purposes, subject to section 19 (1) (a) of the Reserves Act 1977, and further declares the said reserve to form part of the Arnold River Scenic Reserve.

#### Schedule

#### Westland Land District—Grey District

18.4000 hectares, more or less, being Rural Section 6785, situated in Blocks IV and VIII, Hohonu Survey District.

Dated at Wellington this 27th day of July 1991.

W. ROB STOREY, Minister of Lands.

DENIS MARSHALL, Minister of Conservation.

(File Cons. H.O. RES 0969, C.O. 2/06/3/3) ln8087

#### Land Act 1948

#### Reservation of Land

Pursuant to section 167 of the Land Act 1948, the Minister of Conservation, with the consent of the Minister of Lands, hereby sets apart the land described in the Schedule hereto, as a scenic reserve subject to the provisions of section 19 (1) (a) of the Reserves Act 1977.

#### Schedule

#### North Auckland Land District—Auckland City

8093 square metres, more or less, being Allotment 34A, Aotea Parish, situated in Blocks V and VII, Tryphena Survey District. S.O. Plan 2497.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

(Files R.O. DOC 133 H.O. RES. 0156) ln8073

#### Reservation of Land

Pursuant to section 167 of the Land Act 1948, the Minister of Conservation, with the consent of the Minister of Lands, hereby sets apart the land described in the Schedule hereto, as a recreation reserve, subject to the provisions of the Reserves Act 1977.

#### Schedule

#### North Auckland Land District—Papakura District

2381 square metres, more or less, being transferred road adjoining Section 1, S.O. Plan 65144. Shown marked "A" on S.O. Plan 65144. All *New Zealand Gazette*, 1991, page 846.

943 square metres, more or less, being transferred road adjoining Section 1, S.O. Plan 65144. Shown marked "B" on S.O. Plan 65144. All *New Zealand Gazette*, 1990, page 4432.

1.4482 hectares, more or less, being Section 1, S.O. Plan 65144.

All situated in Block VIII, Drury Survey District.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

(File LRC 795) In8074

#### Reservation of Land

Pursuant to section 167 of the Land Act 1948, the Minister of Conservation, with the consent of the Minister of Lands, hereby sets apart the land described in the Schedule hereto, as a recreation reserve, subject to the provisions of the Reserves Act 1977.

#### Schedule

#### North Auckland Land District—Rodney District

5940 square metres, more or less, being Section 1, S.O. Plan 65441, situated in Block XII, Waiwera Survey District. All New Zealand Gazette, 1971, page 2671.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

(Files LRC 792, DOSLI H.O. 90/4)

#### Reservation of Land

Pursuant to section 167 of the Land Act 1948, the Minister of Conservation, with the consent of the Minister of Lands, hereby sets apart the land described in the Schedule hereto, as a local purpose (landscape, carparking) reserve, subject to the provisions of the Reserves Act 1977.

#### Schedule

#### North Auckland Land District—Manukau District

3.6300 hectares, more or less, being Section 1, Block IV, Otahuhu Survey District, S.O. Plan 63292. All *New Zealand Gazette*, 1986, page 4021.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

(File DOC LLP721, DOSLI 91/16) ln8076

#### Reservation of Land

Pursuant to section 167 of the Land Act 1948, the Minister of Conservation, with the consent of the Minister of Lands, hereby sets apart the land described in the Schedule hereto, as a scenic reserve, subject to the provisions of section 19 (1) (a) of the Reserves Act 1977.

#### Schedule

#### North Auckland Land District—Rodney District

2.0432 hectares, more or less, being Section 27, S.O. Plan 59122, situated in Block X, Tauhoa Survey District. Part certificate of title 355/294 canc.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

(File R.O. DOC 269)

#### Reservation of Land

Pursuant to the Land Act 1948, the Minister of Conservation, with the consent of the Minister of Lands, hereby sets apart the land described in the Schedule hereto, as a recreation reserve, subject to the provisions of the Reserves Act 1977.

#### Schedule

#### North Auckland Land District—Waitakere City

6042 square metres, more or less, being Allotment 736, Waikomiti Parish, situated in Block VII, Titirangi Survey

4

District, S.O. Plan 54770. All New Zealand Gazette notice B. 203585.1 (New Zealand Gazette, 1983, page 2393).

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

(Files LRC 765, DOSLI H.O. 90/51)

#### Reservation of Land

Pursuant to section 167 of the Land Act 1948, the Minister of Conservation, with the consent of the Minister of Lands, hereby sets apart the land described in the Schedule hereto, as a scenic reserve, subject to the provisions of section 19 (1) (a) of the Reserves Act 1977.

#### Schedule

#### Nelson Land District—Buller District

5.4983 hectares, more or less, being Sections 45, 48 and 53, Block III, Lewis Survey District.

Dated at Wellington this 27th day of July 1991.

DENIS MARSHALL, Minister of Conservation.

(File Cons. H.O. RES. 0937, C.O. 2.06.2.30) (Lands H.O. Case 91/12)  $_{\mbox{\scriptsize 18079}}$ 

#### Reserves Act 1977

### Vesting of a Reserve in the Rodney District Council

Pursuant to the Reserves Act 1977, and to a delegation from the Minister of Conservation, the Regional Conservator, Auckland Conservancy of the Department of Conservation, hereby vests the recreation reserve described in the Schedule hereto in The Rodney District Council in trust for recreation purposes.

#### Schedule

#### North Auckland Land District

28.8524 hectares, more or less, being Allotments 329, 330 and part Allotments 331 and 332, Mahurangi Parish, situated in Block XII, Mahurangi Survey District, and Block V, Kawau Survey District. Part *New Zealand Gazette*, 1980, page 315. S.O. Plan 32225.

Dated at Auckland this 1st day of August 1991.

G. H. CAMPBELL, Regional Conservator. ln8110

### **Iwi Transition Agency**

#### **Maori Affairs Restructuring Act 1989**

#### Corrigendum

#### Maori Land Development Notice

In the notice issued on the 12th day of July 1991 and published in the *New Zealand Gazette* on the 25th day of July 1991, No. 110, page 2445, amend the description of the land in the Second Schedule from "Mangamuka East 3L2B2" to "Mangamuka West 3L2B2."

Dated at Whangarei this 6th day of August 1991.

R. J. TOPP, for Assistant General Manager.

Iwi Transition Agency (Taitokerau).

(ITA H.O. 15/1/965, 61/3, 61/3A; R.O. 19/L/8) s ln8104

#### Maori Land Development Notice

Pursuant to section 21 of the Maori Affairs Restructuring Act 1989, the Iwi Transition Agency hereby gives notice as follows:

#### Notice

- 1. This notice may be cited as Maori Land Development Notice Gisborne 1991, No. 4.
- 2. The notices referred to in the First Schedule hereto are hereby partially revoked by omitting all reference to the land described in the Second Schedule hereto.
- 3. The land described in the Second Schedule hereto is hereby released from Part II of the Maori Affairs Restructuring Act 1989 (formerly Part XXIV of the Maori Affairs Act 1953).

#### First Schedule

Date of Notice	Reference	Registration No.
2 December 1981	New Zealand Gazette, 10 December 1981, No. 146, page 3736	404241.1
17 March 1972	New Zealand Gazette, 29 March 1972, No. 29, page 691	278418

#### Second Schedule

#### Hawke's Bay Land District

Area
ha
Being
4.1442 Section 1, S.O. 10105, situate in Block VIII,
Mohaka District (T. 258264).

8151 Section 3, S.O. 10105, situate in Block VIII. Mohaka District (T. 258264).

2886 Section 4, S.O. 10105, situate in Mohaka B4 (C.T. M2/381).

8938 Section 6, S.O. 10105, situate in Mohaka B21 (C.T. M2/378).

6590 Section 7, S.O. 10105, situate in Mohaka B24 (C.T. M2/386).

3723 Section 8, S.O. 10105, situate in Mohaka B25 (C.T. M2/387).

Dated at Gisborne this 12th day of July 1991.

For and on behalf of the General Manager, Iwi Transition Agency.

L. R. MOEAU, Assistant General Manager.

Iwi Transition Agency, Gisborne.

#### Maori Land Development Notice

Pursuant to section 21 of the Maori Affairs Restructuring Act 1989, the Iwi Transition Agency hereby gives notice as follows:

#### Notice

- 1. This notice may be cited as Maori Land Development Notice Gisborne 1991, No. 5.
- 2. The land described in the Schedule hereto is hereby declared to be subject to Part II of the Maori Affairs Restructuring Act 1989.

#### Schedule

m<sup>2</sup>

#### Hawke's Bay Land District

Area

5673 Section 2, S.O. 10105, situate in Block IV, Mohaka District (C.T. L2/1254).

Being

ha

3.2078 Being Section 5, S.O. 10105, situate in Mohaka B29X (C.T. L2/1254).

Dated at Gisborne this 12th day of July 1991.

For and on behalf of the General Manager, Iwi Transition Agency.

L. R. MOEAU, Assistant General Manager.

Iwi Transition Agency, Gisborne. ln8108

### \_\_

# New Zealand Railways Corporation

### New Zealand Railways Corporation Act 1981

### Declaring Land at Picton to be Set Apart for Railway Purposes

Pursuant to sections 24 and 30 of the New Zealand Railways Corporation Act 1981 and section 52 of the Public Works Act 1981, the Chief Executive of the New Zealand Railways Corporation hereby declares the land described in the Schedule hereto to be set apart for and on behalf of Her Majesty the Queen for railway purposes.

#### Schedule

#### Marlborough Land District—Marlborough District

All those pieces of land situated in Block XII, Linkwater Survey District described as follows:

Area	
m²	Being
354	Part Section 231, Town of Picton, being part of
(0a0r14p)	the land comprised and described in
	conveyance 5804 (deeds register C4/95).
329	Part Section 232, Town of Picton, being part of
(0a0r13p)	the land comprised and described in
	conveyance 5804 (deeds register C4/95).
316	Part Section 233, Town of Picton, being part of
(0a0r12.5p)	the land comprised and described in
	conveyance 5804 (deeds register C4/95).
227	Part Section 935, Town of Picton, being part of
(0a0r9p)	the land comprised and described in
	conveyance 5804 (deeds register C4/95).
1011	Section 963, Town of Picton, being balance of
(0a1r0p)	the land comprised and described in
	conveyance 5804 (deeds register C4/95).

Dated at Wellington this 13th day of August 1991.

P. K. TROTMAN, for Chief Executive, New Zealand Railways Corporation.

(NZR L.O. 21188/-) ln8112

### Declaring Land at Picton to be Set Apart for Railway Purposes

Pursuant to sections 24 and 30 of the New Zealand Railways Corporation Act 1981 and section 52 of the Public Works Act 1981, the Chief Executive of the New Zealand Railways Corporation hereby declares the land described in the Schedule hereto to be set apart for and on behalf of Her Majesty the Queen for railway purposes.

#### Schedule

#### Marlborough Land District—Marlborough District

All those pieces of land situated in Block XII, Linkwater Survey District described as follows:

Area	_
$m^2$	Being
253	Part Section 230, Town of Picton, being part of
(0a0r10p)	the land comprised and described in
	conveyance 5270 (deeds register C3/618).
341	Part Section 234, Town of Picton, being part of
(0a0r13.5p)	the land comprised and described in
	conveyance 5270 (deeds register C3/618).
379	Part Section 235, Town of Picton, being balance
(0a0r15p)	of the land comprised and described in
	conveyance $5270$ (deeds register $C3/618$ ).

Dated at Wellington this 13th day of August 1991.

 $\mbox{P. K. TROTMAN, for Chief Executive, New Zealand Railways Corporation.}$ 

(NZR L.O. 21188/-) in8115

### Declaring Land at Picton to be Set Apart for Railway Purposes

Pursuant to sections 24 and 30 of the New Zealand Railways Corporation Act 1981 and section 52 of the Public Works Act 1981, the Chief Executive of the New Zealand Railways Corporation hereby declares the land described in the Schedule hereto to be set apart for and on behalf of Her Majesty the Queen for railway purposes.

#### Schedule

#### Marlborough Land District—Marlborough District

All that piece of land containing 265 square metres (0a0r10.5p), situated in Block XII, Linkwater Survey District, being part Section 237, Town of Picton, and being all of the land comprised and described in conveyance 5527 (deeds index 1/280, deeds index 1/282).

Dated at Wellington this 13th day of August 1991.

 $P.\ K.\ TROTMAN,$  for Chief Executive, New Zealand Railways Corporation.

(NZR L.O. 21188/-) ln8114

### Declaring Land at Helensville to be Acquired for Railway Purposes

Pursuant to sections 24 and 30 of the New Zealand Railways Corporation Act 1981 and section 20 of the Public Works Act 1981, the Chief Executive of the New Zealand Railways Corporation hereby declares that, an agreement to that effect having been entered into, the land described in the Schedule hereto is hereby acquired for and on behalf of Her Majesty the Queen for railway purposes.

#### Schedule

#### North Auckland Land District—Rodney District

Both those pieces of land situated in Block XIV, Kaipara Survey District described as follows:

Area m<sup>2</sup> Being 905 Part Lot 7, D.P. 45027; marked 'A' on plan. 512 Part Lot 1, D.P. 109291; marked 'B' on plan.

As shown marked as above mentioned on Plan L.O. 36300 (S.O. 65373), lodged in the office of the New Zealand Railways Corporation at Wellington.

Dated at Wellington this  $12th\ day\ of\ August\ 1991.$ 

P. K. TROTMAN, for Chief Executive, New Zealand Railways Corporation.

(NZR L.O. 32942/13) ln8116

### Declaring Land at Picton to be Set Apart for Railway Purposes

Pursuant to sections 24 and 30 of the New Zealand Railways Corporation Act 1981 and section 52 of the Public Works Act 1981, the Chief Executive of the New Zealand Railways Corporation hereby declares the land described in the Schedule hereto to be set apart for and on behalf of Her Majesty the Queen for railway purposes.

#### Schedule

#### Marlborough Land District—Marlborough District

All that piece of land containing 1011 square metres (0a1r0p), situated in Block XII, Linkwater Survey District, being Section 236, Town of Picton and being all of the land comprised and described in conveyance 5266 (deeds index 1/840).

Dated at Wellington this 13th day of August 1991.

 $P.\ K.\ TROTMAN,$  for Chief Executive, New Zealand Railways Corporation.

(NZR L.O. 21188/-)

### **Survey and Land Information**

#### **Public Works Act 1981**

#### Road Stopped in the City of Nelson

Pursuant to section 116 (1) of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Nelson, declares the portion of road described in the Schedule hereto to be stopped.

#### Schedule

#### Nelson Land District—Nelson City

80 square metres, situated in Block XVI, Moutere Survey District, adjoining or passing through Lot 1, D.P. 3033; as shown marked "A" on S.O. Plan 14651, lodged in the office of the Chief Surveyor at Nelson.

Dated at Nelson this 5th day of August 1991.

R. G. C. WRATT, Manager, Lands and Property.

(DOSLI Nn. D.O. 72/6/11/2/0/77) ln8105

1CL

#### Road Acquired for Road in the City of Nelson

Pursuant to section 20 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Manager, Lands and Property, Department of Survey and Land Information, Nelson, declares that, an agreement to that effect having been entered into, the land described in the Schedule hereto is hereby acquired for road and shall vest in The Nelson City Council on the date of publication hereof in the *Gazette*.

#### Schedule

#### Nelson Land District—Nelson City

All those pieces of land situated in Block IV, Waimea Survey District, described as follows:

Area m<sup>2</sup>

2

5 Part Section 444, City of Nelson; marked "A" on plan.

Being

- 76 Part Section 444, City of Nelson; marked "B" on plan.
- 71 Part Lot 6, D.P. 1537; marked "C" on plan.

As shown marked as above mentioned on S.O. Plan 14725, lodged in the office of the Chief Surveyor at Nelson.

Dated at Nelson this 5th day of August 1991.

R. G. C. WRATT, Manager, Lands and Property.

(DOSLI Nn. D.O. 7975-03552)

1CL

### Land in Thames-Coromandel District Declared to be Road

Pursuant to section 114 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager of the Department of Survey and Land Information, Hamilton, declares the land described in the Schedule hereto to be road.

#### Schedule

#### South Auckland Land District

Area m²

Being

1173 Part Crown land reserved under Section 129, Land Act 1924; marked "A" on S.O. Plan 56916.

Situated in Block I, Whitianga Survey District.

- 24 Part Crown land, tramway reserve; marked "D" on S.O. Plan 56915.
- 13 Part Crown land, old bed Owhero Creek; marked "D" on S.O. Plan 56914.
- 7 Part Crown land, reserved under section 110, Land Act 1892; marked "E" on S.O. Plan 56914.

Situated in Block X, Otama Survey District.

704 Part Crown land; marked "C" on S.O. Plan 56914. Situated in Block X, Otama Survey District and Block XVI, Coromandel Survey District.

274 Part Crown land; marked "B" on S.O. Plan 56914. Situated in Block XVI, Coromandel Survey District.

As shown on the plans marked as above mentioned and lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 31st day of July 1991.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 72/25/2C/06/77) ln8099

1CL

### Amending a Declaration Acquiring Land in Thames-Coromandel District for Road

Pursuant to section 55 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor of the Department of Survey and Land Information, Hamilton, hereby amends the declaration dated the 14th day of June 1991 and published in the *New Zealand Gazette* of 20 June 1991, No. 90, page 1974, by deleting the land firstly described in the Schedule.

Dated at Hamilton this 31st day of July 1991.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 72/25A/2C/01/1)

1CL

#### Land in Thames-Coromandel District Acquired for Road and for the Functioning Indirectly of a Road

Pursuant to section 20 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the District Solicitor of the Department of Survey and Land Information, Hamilton, declares that, an agreement to that effect having been entered into, the land described in the First Schedule hereto is hereby acquired for road and the land described in the Second Schedule hereto is hereby acquired for the functioning indirectly of a road, and further declares that the land in both Schedules shall vest in the Crown on the date of publication hereof in the Gazette.

## 15 AUGUST First Schedule

#### South Auckland Land District

Area m² Being
747 Part Lot 1, D.P. S. 55784; marked "B" on plan.
831 Part Lot 1, D.P. S. 55784; marked "C" on plan.
153 Part Lot 3, D.P. S. 55784; marked "D" on plan.

Situated in Block IV, Ohinemuri Survey District.

As shown marked as above mentioned on S.O. Plan 58840, lodged in the office of the Chief Surveyor at Hamilton.

#### Second Schedule

#### South Auckland Land District

1.3608 hectares, situated in Block IV, Ohinemuri Survey District, being part Lot 1, D.P. S. 55784; as shown marked "A" on S.O. Plan 58840, lodged in the office of the Chief Surveyor at Hamilton.

Dated at Hamilton this 30th day of July 1991.

W. G. KORVER, District Solicitor.

(DOSLI Hn. 72/25/2C/08/12) in8096

1CL

1CL

### Stopped Road in Thames-Coromandel District Vested

Pursuant to section 117 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager of the Department of Survey and Land Information, Hamilton, declares that the pieces of stopped road described in the Schedule hereto shall be amalgamated with the land in certificate of title, Volume 1045, folio 111, subject to memorandum of mortgage H. 993524.3, South Auckland Land Registry.

#### Schedule

#### South Auckland Land District

Area

m<sup>2</sup> Being

Situated in Blocks I and IV, Whitianga Survey District.

572 Section 2, S.O. 56923.

2375 Section 1, S.O. 56922.

Situated in Block I, Whitianga Survey District.

Dated at Hamilton this 30th day of July 1991.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 72/25/2C/06/79)

0/79)

In8097

### Stopped Road in Thames-Coromandel District Vested

Pursuant to section 117 of the Public Works Act 1981, and to a delegation from the Minister of Lands, the Assistant District Manager of the Department of Survey and Land Information, Hamilton, declares that the piece of stopped road described in the Schedule hereto shall be amalgamated with the land in certificate of title No. 32C/312, South Auckland Land Registry.

#### Schedule

#### South Auckland Land District

388 square metres, situated in Block IV, Whitianga Survey District, being Section 1, S.O. 56558.

Dated at Hamilton this 31st day of July 1991.

R. W. BARNABY, Assistant District Manager.

(DOSLI Hn. 72/25/2C/06/67) ln8098

1CL

### Treasury

#### State-Owned Enterprises Act 1986

### The State-Owned Enterprises (Landcorp Farming Limited Vesting Order No. 10) Order 1991

CATHERINE A. TIZARD, Governor-General ORDER IN COUNCIL

At Wellington this 12th day of August 1991

Present

HER EXCELLENCY THE GOVERNOR-GENERAL IN COUNCIL

Pursuant to sections 24 (1) and 28 of the State-Owned Enterprises Act 1986, His Excellency the Governor-General, acting by and with the advice and consent of the Executive Council, hereby makes the following order.

#### Order

- 1. Title—This order may be cited as the State-Owned Enterprises (Landcorp Farming Limited Vesting Order No. 10) Order 1991.
- **2. Application**—This order applies to all the land described in the Schedule to this order, saving and excepting—
  - (a) The bed of any navigable river (within the meaning of section 261 of the Coal Mines Act 1979); and
  - (b) The bed of any river or stream that has an average width of 3 metres or more; and
  - (c) The bed of any lake that has an area (calculated on the basis of its normal level or maximum control level and the normal level or maximum control level of all its bays and inlets) of more than 8 hectares; and
  - (d) The bed of any bay or inlet of any such lake.
- 3. Approval and vesting—The land to which this order applies—
  - (a) Is hereby approved for the purposes of section 24 (1) (b) of the State-Owned Enterprises Act 1986; and
  - (b) On the 12th day of September 1991 shall vest for an estate in fee simple in Landcorp Farming Limited (a company duly incorporated under the Companies Act 1955, having its registered office at Wellington, and being a State enterprise within the meaning of section 29 (1) of the State-Owned Enterprises Act 1986),—
    - (i) Together with the appurtenant rights (if any) described in the Schedule to this order; but
    - (ii) Subject to the rights, interests, and encumbrances, (if any) to which it is described in the Schedule to this order as being subject.

Schedule Cl. 2

#### Legal Description of Land

Area Description

North Auckland Land District

4.3238 ha Part Section 7S, Puketi Settlement, situated in Block XV, Kaeo Survey District, and shown on S.O. Plan 45086.

Taranaki Land District

2437.8696 ha The following land:

(1) Section 1, S.O. Plan 13227, situated in Blocks I and II, Tainui Survey District.

Area Description Area Description (2) Sections 1 and 2, S.O. Plan 13236, (2) All the land situated in Block V, Mahoe situated in Block IV, Tainui Survey District Survey District, shown marked "A" "C", "D", and "E" on S.O. Plan 13283. and Block VII, Mokau Survey District. (3) Section 1, S.O. Plan 13221, situated in (3) All the land situated in Block II, Mahoe Survey District, shown marked "A", "B", "C", "D", "E", "F", "G", "H", "I", and "J" on S.O. Plan 13284. Blocks I and III, Tainui Survey District. (4) Section 1, S.O. Plan 13220, situated in Blocks II and IV, Tainui Survey District. (5) Section 1, S.O. Plan 13222, situated in (4) Section 29, Block I, Mahoe Survey Blocks I, III, and IV, Tainui Survey District, shown on S.O. Plan 4792. (5) Section 13, Block V, Mahoe Survey District, shown on S.O. Plan 3086. (6) Lot 4, D.P. 3174, situated in Blocks III and IV, Tainui Survey District. (6) Sections 1 and 2, S.O. Plan 13206, situated in Block XIII, Pouatu Survey (7) Mokau-Mohakatino 1A2, situated in Block IV, Tainui Survey District, and District and Block I, Mahoe Survey shown on M.L. Plan 605. District. (8) Section 4, Block IV, Tainui Survey District, shown on M.L. Plan 605. Wellington Land District Together with Roadway created by a 2526.9 ha The land situated in Blocks XII and XVI, Maori Land Court Order being Document Kaitieke Survey District, Blocks IX and No. 146805 (Taranaki Registry). Subject XIII, Tongariro Survey District, and Block to gas pipeline easements contained in Easement Certificates Nos. 165786, I, Ruapehu Survey District, shown marked "A", "B", "C", and "D" on S.O. Plan 269146, and 269147 (Taranaki Registry). 36609. 1521.5833 ha The following land: Subject to rights of way reserved over parts (1) Section 1, S.O. Plan 13228, situated in of the land shown marked "A" on S.O. Plan 36609, shown marked "E", "F", and Block III, Upper Waitara Survey District. (2) Sections 1 and 2, S.O. Plan 13231, "G" on L.T. Plan 72594, and shown marked "H" on L.T Plan 72595, in favour situated in Block III, Upper Waitara Survey District. of the Minister of Conservation. (3) Section 1, S.O. Plan 13232, situated in Together with rights of way over Parts Block XII, Mimi Survey District. Waimarino No. 1, shown marked "B" on (4) Section 1, S.O. Plan 13229, situated in L.T. Plan 72592, marked "C" on L.T Plan 72593 and marked "D" on L.T. Plan Block IV, Upper Waitara Survey District. (5) Sections 1, 2, and 3, S.O. Plan 13219, situated in Block III, Upper Waitara Together with right of way over Part Section Survey District. 21, Block XII, Kaitieke Survey District, shown marked "A" on L.T. Plan 72592. (6) Section 1, S.O. Plan 13218, situated in Block III, Upper Waitara Survey District Together with right of way over Part and Sections 2, 3, and 4, S.O. Plan 13218, Waimarino No. 1, being a marginal strip situated in Block XII, Mimi Survey pursuant to section 24 (3) of the Conservation Act 1987, shown marked (7) Section 1, S.O. Plan 13211, situated in "J" on L.T. Plan 72592. Block III, Upper Waitara Survey District. 2168 m<sup>2</sup> Section 131, Motoa District, situated in (8) Section 1, S.O. Plan 13233, situated in Block XI, Mount Robinson Survey District, Block IV, Upper Waitara Survey District. and shown on S.O. Plan 34371. (9) All the land situated in Block III, Upper 1438.9171 ha Section 1, S.O. Plan 36557, Section 1, S.O. Waitara Survey District, shown marked "A", "B", "C", "D", "E", "G", and "H" Plan 36558, Section 1, S.O. Plan 36559, Sections 1 and 2, S.O. Plan 36566, Sections 1 and 2, S.O. Plan 36567, and on S.O. Plan 13277. (10) All the land situated in Block IV, Upper Section 1, S.O. Plan 36568, situated in Waitara Survey District, shown marked Blocks IV, VI, IX and X, Mount Robinson "A" and "B" on S.O. Plan 13282. Survey District. The land situated in Block XIV, Wairarapa 497.40 ha The following land: 1432.544 ha Survey District, shown marked "A", "B", (1) All the land situated in Blocks XII and "C" and "D" on S.O. Plan 36605, and XIII, Mimi Survey District, Block I, Pouatu marked "A" on S.O. Plan 36606. Survey District, Block XIII, Waro Survey Lot 1, D.P. 25682, situated in Block XIV, 8.3317 ha District, and Block IV, Upper Waitara Survey District, shown marked "A", "B", "C", "D", "E", "F", "G", "I", "J", "K", "L", "M", "N", "O", "P", "Q", "R", "S", "T", "U", "V", "W", "X", "Y", "Z", and "AA" on S.O. Plan 13276. Wairarapa Survey District. Sections 1, 2, 3, and 4, S.O. Plan 36541, situated in Block I, Waiopehu Survey District and Block II, Waitohu Survey 103.3940 ha District. Subject to gas pipeline easement created by Document 754568 (Wellington (2) Section 1, S.O. Plan 13252, situated in Registry) and Proclamation 500359 Block I, Pouatu Survey District. (Wellington Registry) defining the middle (3) Section 2, S.O. Plan 13252, situated in line of a motorway. Block IV, Upper Waitara Survey District. 647.3250 ha Section 1, S.O. Plan 36550, situated in 1282.2103 ha The following land: Block XI, Puketi Survey District and Block (1) All the land situated in Blocks I and II, IV, Pihanga Survey District.

Nelson Land District

Mahoe Survey District, shown marked

"A" on S.O. Plan 13281.

13 AUGUS1	NEW ZEALAN	D GAZLIIL	2073
Area	Description	Area	Description
561.6630 ha	Sections 1, 2, 3, 4, and 5, S.O. Plan 14711, situated in Blocks IV and VIII, Aorere Survey District. Subject to Prospecting Licence Volume 8D, folio 71 (Nelson Registry).		<ul> <li>(3) Sections 3 and 5, S.O. Plan 23490, situated in Block I, Waipori Survey District.</li> <li>(4) Section 4, S.O. Plan 23490, situated in Block IV, Hedgehope Survey District,</li> </ul>
125.2465 ha	Sections 1, 2, 3, and 4, S.O. Plan 14709, situated in Blocks XIV and XV, Pakawau Survey District.		Block XIV, Tuapeka East Survey District, and Block I, Waipori Survey District. (5) Section 6, S.O. Plan 23490, situated in
1350.7250 ha	Sections 1, 2, and 3, S.O. Plan 14710, situated in Blocks X, XI, XIV, and XV, Motueka Survey District. Subject to right of way created by document 6D/974 (Nelson Registry), and subject to right of way created by document 53905 (Nelson	7803.582 ha	Block XIV, Tuapeka East Survey District and Block I, Waipori Survey District.  (6) Section 7, S.O. Plan 23490, situated in Blocks I and IV, Waipori Survey District and Blocks II and VII, Town of Waipori.
	Registry).	7003.302 Ha	The following land:
660.4800 ha	Section 1, S.O. Plan 12644, situated in Blocks III, IV, VII, and VIII, Waimea Survey District, and Section 1 S.O. Plan 14698, situated in Blocks IV and VIII, Waimea Survey District.		<ol> <li>Section 1, S.O. Plan 23526, situated in Block II, Hedgehope Survey District and Blocks XV and XVI, Tuapeka East Survey District.</li> <li>Sections 2 and 9, S.O. Plan 23526, situated in Blocks XV and XVI, Tuapeka</li> </ol>
	Canterbury Land District		East Survey District. (3) Section 3, S.O. Plan 23526, situated in
183.3781 ha	The following land: (1) Section 1, S.O. Plan 9633, situated in Block II, Lyndon Survey District.		Blocks II and III, Hedgehope Survey District and Blocks XIV and XV, Tuapeka East Survey District.
	(2) Rural Section 38211, situated in Block II, Lyndon Survey District, and shown on S.O. Plan 9231.		(4) Section 4, S.O. Plan 23526, situated in Blocks XIV and XV, Tuapeka East Survey District.
	(3) Rural Section 40031, situated in Block II, Lyndon Survey District, and shown on S.O. Plan 11745.		(5) Sections 5 and 19, S.O. Plan 23526, situated in Block XIV, Tuapeka East Survey District.
	(4) Sections 1, 2, 3, 4, 5, 6, and 7, S.O. Plan 18605, situated in Block II, Lyndon Survey District.		(6) Section 7, S.O. Plan 23526, situated in Blocks III and IV, Hedgehope Survey District and Block XIV, Tuapeka East Survey District.
	Otago Land District		(7) Section 8, S.O. Plan 23526, situated in
617.4488 ha	The following land: (1) Section 1, S.O. Plan 21695, situated in		Block IV, Hedgehope Survey District and Block XIV, Tuapeka East Survey District.
	Block XIV, Clutha Survey District. (2) Sections 1 and 2, S.O. Plan 21691,		(8) Section 10, S.O. Plan 23526, situated in Blocks XII, XIV, and XV, Tuapeka East
	situated in Block XV, Clutha Survey		Survey District. (9) Section 11, S.O. Plan 23526, situated in
	District. (3) Section 1, S.O. Plan 23473, situated in		Blocks XI, XII, XIV, and XV, Tuapeka East Survey District.
	Blocks XV and XVI, Clutha Survey District.		(10) Sections 12 and 13, S.O. Plan 23526, situated in Blocks XI and XIV, Tuapeka
	(4) Section 1, S.O. Plan 23474, situated in Blocks XII, XV and XVI, Clutha Survey		East Survey District.
	District. (5) Sections 1 and 2, S.O. Plan 23484,		(11) Sections 14, 15, 16, 17, and 18, S.O. Plan 23526, situated in Block IV, Waipori
	situated in Blocks VI and X, Clutha Survey District.		Survey District. (12) Section 20, S.O. Plan 23526, situated
	(6) Sections 1, 2 and 5, S.O. Plan 23485, situated in Blocks X and XI, Clutha		in Block II, Hedgehope Survey District and Block XV, Tuapeka East Survey
	Survey District. (7) Sections 1 and 2, S.O. Plan 23486, situated in Blocks VI, X and XI, Clutha		District. Subject (in relation to all 7803.582 ha) to Mining Licence Volume 9D, folio 134
	Survey District. (8) Sections 1, 3, 4, and 5, S.O. Plan 23487, situated in Blocks X and XIV, Clutha Survey District.	1.4418 ha	(Otago Registry).  All the land situated in Blocks XIV and XV, Town of Blueskin shown marked "A" on S.O. Plan 23553.
6592 m²	Section 2, S.O. Plan 23487, situated in Block XIV, Clutha Survey District.	MARIE SHROF	F, Clerk of the Executive Council.
2232.0360 ha	The following land:	Explanatory N	Vote
	(1) Section 1, S.O. Plan 23490, situated in Blocks VIII and X, Lee Stream Survey	This note is not p general effect.	part of the order, but is intended to indicate its
	District and Blocks I and II, Waipori Survey District.  (2) Sections 2 and 8, S.O. Plan 23490, situated in Blocks I and IV, Waipori Survey District.		s certain Crown land in Landcorp Farming diary of a State enterprise, on 12 September

Survey District.

#### The State-Owned Enterprises (Landcorp Management Services Limited Vesting Order No. 12) Order 1991

### CATHERINE A. TIZARD, Governor-General ORDER IN COUNCIL

At Wellington this 12th day of August 1991

#### Present:

HER EXCELLENCY THE GOVERNOR-GENERAL IN COUNCIL

Pursuant to sections 24 (1) and 28 of the State-Owned Enterprises Act 1986, His Excellency the Governor-General, acting by and with the advice and consent of the Executive Council, hereby makes the following order.

#### Order

- 1. Title—This order may be cited as the State-Owned Enterprises (Landcorp Management Services Limited Vesting Order No. 12) Order 1991.
- **2. Application**—This order applies to all the land described in the Schedule to this order, saving and excepting—
  - (a) The bed of any navigable river (within the meaning of section 261 of the Coal Mines Act 1979); and
  - (b) The bed of any river or stream that has an average width of 3 metres or more; and
  - (c) The bed of any lake that has an area (calculated on the basis of its normal level or maximum control level and the normal level or maximum control level of all its bays and inlets) of more than 8 hectares; and
  - (d) The bed of any bay or inlet of any such lake.
- **3. Approval and vesting**—The land to which this order applies—
  - (a) Is hereby approved for the purposes of section 24 (1) (b) of the State-Owned Enterprises Act 1986; and
  - (b) On the 12th day of September 1991 shall vest for an estate in fee simple in Landcorp Management Services Limited (a company duly incorporated under the Companies Act 1955, having its registered office at Wellington, and being a State enterprise within the meaning of section 29 (1) of the State-Owned Enterprises Act 1986),—
    - (i) Together with the appurtenant rights (if any) described in the Schedule to this order; but
    - (ii) Subject to the rights, interests, and encumbrances, (if any) to which it is described in the Schedule to this order as being subject.

#### Schedule Cl. 2

#### Legal Description of Land

cgai Decemp	
Area	Description
	North Auckland Land District
2690 m²	Section 1, S.O. Plan 64296, situated in Block III, Kaeo Survey District.
4.4605 ha	Sections 1 and 2, S.O. Plan 65591, situated in Block XIV, Tauhoa Survey District.
17.0727 ha	Section 38, Block VI, Pakiri Survey District, shown on S.O. Plan 42883.
960 m²	Section 1, S.O. Plan 64262, situated in Block I, Kaeo Survey District.
	South Auckland Land District
3111 m²	Section 1, S.O. Plan 41916, situated in Block VII, Piako Survey District.
1219 m²	Section 11, Block V, Maungatautari Survey District, shown on S.O. Plan 33510.
1.4428 ha	The following land: (1) Section 1, S.O. Plan 58768, situated in Block III, Marotiri Survey District:

Area	Description				
	(2) Section 32, Block III, Marotiri Surv	/ey			
	District shown on S.O. Plan 41416				

#### Taranaki Land District

- 4048 m<sup>2</sup> Sections 6, 7, 8, and 9, Puniwhakau Village, situated in Block III, Omoana Survey District, and shown on S.O. Plan 7770.
- 1.4620 ha Section 1, S.O. Plan 13240, situated in Block V, Ohura Survey District.

#### Wellington Land District

- 1.3628 ha Lot 9, D.P. 28407, situated in Block III, Pihanga Survey District.
  - 3741 m<sup>2</sup> Section 152, Mananuni Village, situated in Block II, Hunua Survey District, and shown on S.O. Plan 26415.
  - 416 m<sup>2</sup> Section 14, Town District, situated in Block X, Port Nicholson Survey District, and shown on S.O. Plan 23396. Subject to lease Vol. 699, folio 89 (Wellington Registry).
- 2982 m<sup>2</sup> Section 25, Block XIV, Belmont Survey
  District, shown on S.O. Plan 23969.
  Subject to lease Vol. 905, folio 24
  (Wellington Registry).
- 7532 m<sup>2</sup> Section 24, Block XIV, Belmont Survey
  District, shown on S.O. Plan 23969.
  Subject to lease Vol. 805, folio 65
  (Wellington Registry).
- 7.1943 ha Section 1, Block XIV, Belmont Survey
  District, shown on S.O. Plan 21701.
  Subject to lease Vol. 805, folio 65
  (Wellington Registry).
  - 946 m<sup>2</sup> Section 62, Block XIV, Belmont Survey District, shown on S.O. Plan 25895. Subject to lease Vol. 805, folio 65 (Wellington Registry).
- 22.9930 ha Section 1, S.O. Plan 36521, situated in Blocks XIV and XV, Belmont Survey District.
  - 5782 m<sup>2</sup> Section 2, S.O. Plan 36521, situated in Blocks XIV and XV, Belmont Survey District.
  - 220 m<sup>2</sup> Section 1, S.O. Plan 36552, situated in Block VII, Port Nicholson Survey District.
  - 1299 m<sup>2</sup> Sections 1 and 2, S.O. Plan 36551, situated in Block III, Mount Adams Survey District.

#### Westland Land District

- 1.2170 ha Section 1, S.O. Plan 11730, situated in Block XV, Arnold Survey District.
  - 9760 m<sup>2</sup> Section 2, S.O. Plan 11730, situated in Block II, Kopara Survey District.
  - 2023 m<sup>2</sup> Section 1, S.O. Plan 11725, situated in Block I, Toaroha Survey District.
  - 1890 m<sup>2</sup> Section 2, S.O. Plan 11725, situated in Block I, Toaroha Survey District.
  - 2023 m<sup>2</sup> Section 3, S.O. Plan 11725, situated in Block I, Toaroha Survey District.
  - 2223 m<sup>2</sup> Section 4, S.O. Plan 11725, situated in Block I, Toaroha Survey District.
  - 2023 m<sup>2</sup> Section 5, S.O. Plan 11725, situated in Block I, Toaroha Survey District.
- 2023 m<sup>2</sup> Section 6, S.O. Plan 11725, situated in Block I, Toaroha Survey District.
- 3280 m<sup>2</sup> Section 1, S.O. Plan 11722, situated in Block XVI, Mahinapua Survey District.
- 4250 m<sup>2</sup> Section 2, S.O. Plan 11722, situated in Block I, Toaroha Survey District.
- 3560 m<sup>2</sup> Section 1, S.O. Plan 11711, situated in Block V, Toaroha Survey District.

Area	Description	Area	Description
2023 m <sup>2</sup>	Section 2, S.O. Plan 11711, situated in	4997 m <sup>2</sup>	Section 33, Block X, Hanmer Township,
1.1840 ha	Block VI, Toaroha Survey District. Section 2, S.O. Plan 11704, situated in		situated in Block II, Lyndon Survey District, and shown on S.O. Plan 16053.
4960 m²	Block XI, Whataroa Survey District.  Section 1, S.O. Plan 11706, situated in		Subject to Proclamation No. 555510 (Canterbury Registry) declaring the land
12.9400 ha	Block XIII, Hapuka Survey District. Section 1, S.O. Plan 11721, situated in	809 m²	to be a Thermal Water Area.  Section 4, Block X, Hanmer Township,
4050 m²	Block I, Poerua Survey District. Section 1, S.O. Plan 11713, situated in		situated in Block II, Lyndon Survey District, and shown on S.O. Plan 7821.
8.5050 ha	Block IV, Mahinapua Survey District. Section 1, S.O. Plan 11712, situated in Block XIII, Kaniere Survey District and		Subject to Proclamation No. 555510 (Canterbury Registry) declaring the land to be a Thermal Water Area.
	Block Alli, Namere Survey District and Block I, Toaroha Survey District. Subject to Mining Licence registered as Volume 6C, folio 19 (Westland Registry).	809 m²	Section 29, Block XIV, Hanmer Township, situated in Block I, Lyndon Survey District, and shown on S.O. Plan 6594.
4.8960 ha	Sections 2 and 3, S.O. Plan 11712, situated in Block XIII, Kaniere Survey District and Block I, Toaroha Survey District.		Subject to Proclamation No. 555510 (Canterbury Registry) declaring the land to be a Thermal Water Area.
2735 m²	Sections 1 and 2, S.O. Plan 11708, situated in Block X, Waimea Survey District.	809 m²	Section 30, Block XIV, Hanmer Township, situated in Block I, Lyndon Survey
15.3020 ha	Section 1, S.O. Plan 11704, situated in Block XI, Whataroa Survey District.		District, and shown on S.O. Plan 6594. Subject to Proclamation No. 555510
2.6485 ha	Section 1, S.O. Plan 11703, situated in Block IV, Brunner Survey District. Subject		(Canterbury Registry) declaring the land to be a Thermal Water Area.
	to right to drain sewage contained in Register Book Volume 5B, folio 810 (Westland Registry).	809 m²	Section 31, Block XIV, Hanmer Township, situated in Block I, Lyndon Survey District, and shown on S.O. Plan 6594.
$2023\ m^2$	Section 2, S.O. Plan 11706, situated in Block XIII, Hapuka Survey District.		Subject to Proclamation No. 555510 (Canterbury Registry) declaring the land
8550 m²	Section 3, S.O. Plan 11706, situated in Block XIII, Hapuka Survey District.	5.8376 ha	to be a Thermal Water Area.  Lot 3, D.P. 1522, situated in Block XII,
783 m²	Section 1, S.O. Plan 11707, situated in Block VII, Mahinapua Survey District.	50.1700 ha	Nimrod Survey District. Section 1, S.O. Plan 18564, situated in
1820 m²	Subject to Mining Licence registered as Volume 7A, folio 104 (Westland Registry).  Section 1, S.O. Plan 11709, situated in	1865 m²	Block XII, Nimrod Survey District.  Section 2, S.O. Plan 18564, situated in Block XII, Nimrod Survey District.
1020 III	Block XIV, Waimea Survey District.	6.0500 ha	Rural Section 41883, situated in Block XII, Nimrod Survey District, and shown on S.O. Plan 16193.
22.2880 ha	Canterbury Land District Section 76, Hanmer Town Area, situated in	266.8500 ha	Rural Section 41884, situated in Blocks XII
22.2000 Ha	Block I, Lyndon Survey District, and shown on S.O. Plan 11847. Subject to	580 1300 ha	and XVI, Nimrod Survey District, and shown on S.O. Plan 16193. Rural Section 41885, situated in Blocks XII
	Proclamation No. 555510 (Canterbury Registry) declaring the land to be a		and XVI, Nimrod Survey District, and shown on S.O. Plan 16193.
12.9750 ha	Thermal Water Area.  Section 1, S.O. Plan 18574, situated in Blocks I and II, Lyndon Survey District.	28.5303 ha	Rural Section 28855, situated in Block XII, Nimrod Survey District, and shown on S.O. Plan 3501.
	Subject to Proclamation No. 555510 (Canterbury Registry) declaring the land	1873 m²	Section 1, S.O. Plan 18579, situated in Block XIII, Tekapo Survey District.
25.0730 ha	to be a Thermal Water Area. Section 2, S.O. Plan 18574, situated in	1268 m²	Section 2, S.O. Plan 18579, situated in Block XIII, Tekapo Survey District.
	Blocks I and II, Lyndon Survey District. Subject to Proclamation No. 555510 (Canterbury Registry) declaring the land	6468 m² 1144 m²	Section 1, S.O. Plan 18562, situated in Block XV, Rangiora Survey District. Section 2, S.O. Plan 18562, situated in
1.9822 ha	to be a Thermal Water Area. Section 1, S.O. Plan 18573, situated in	631 m²	Block XV, Rangiora Survey District. Section 1, S.O. Plan 18539, situated in
	Blocks I and II, Lyndon Survey District. Subject to Proclamation No. 555510	6020 m²	Block X, Oxford Survey District. Section 1, S.O. Plan 18532, situated in
	(Canterbury Registry) declaring the land to be a Thermal Water Area.	3280 m²	Block II, Arowhenua Survey District. Section 2, S.O. Plan 18532, situated in
531 m²	Section 32, S.O. Plan 18573, situated in Block II, Lyndon Survey District. Subject to Proclamation No. 555510 (Canterbury Registry), declaring the land to be a	1.2140 ha	Block II, Arowhenua Survey District. Rural Section 35563, situated in Block XII, Nimrod Survey District, and shown on
7880 m²	Registry) declaring the land to be a Thermal Water Area.  Section 2, Block X, Hanmer Township,	1193 m²	S.O. Plan 4553.  Section 1, S.O. Plan 18563, situated in Blocks X and XIV, Coleridge Survey
	situated in Blocks I and II, Lyndon Survey District, and shown on S.O. Plan 16053.	4922 m²	District. Section 2, S.O. Plan 18563, situated in
	Subject to Proclamation No. 555510 (Canterbury Registry) declaring the land to be a Thermal Water Area.	1.2187 ha	Block XIV, Coleridge Survey District.  Section 1, S.O. Plan 18578, situated in Block IX, Selwyn Survey District.
	55 W MIGHINI TTUID MIGU.		2.001 III, Octobril Outvey District.

#### **NEW ZEALAND GAZETTE**

Description	Area	Description
Southland Land District	708 m <sup>2</sup>	Section 122, Block IX, Town of Manapouri,
Lot 1, D.P. 10441, situated in Block I,	9 5847 ha	shown on S.O. Plan 9043. Section 10, Block XII, Hauroko Survey
Section 1, S.O. Plan 11736, situated in	9. <b>3047</b> Ha	District, shown on S.O. Plan 7482.
Block I, Manapouri Survey District.	MARIE SHROFE	F, Clerk of the Executive Council.
,		,
	Explanatory N	lote
Manapouri, shown on S.O. Plan 5980.	-	part of the order, but is intended to indicate its
Section 116, Block IX, Town of Manapouri,	general effect.	
	This order vests of	certain Crown land in Landcorp Management
Section 118, Block IX, Town of Manapouri, shown on S.O. Plan 9043.	Services Limited	, a subsidiary of a State enterprise, on 12
Section 119, Block IX, Town of Manapouri, shown on S.O. Plan 9043.	ln8057	•
	Southland Land District  Lot 1, D.P. 10441, situated in Block I, Manapouri Survey District.  Section 1, S.O. Plan 11736, situated in Block I, Manapouri Survey District.  Section 2, S.O. Plan 11736, situated in Block I, Manapouri Survey District.  Part Section 114, Block IX, Town of Manapouri, shown on S.O. Plan 5980.  Section 116, Block IX, Town of Manapouri, shown on S.O. Plan 9043.  Section 118, Block IX, Town of Manapouri, shown on S.O. Plan 9043.  Section 119, Block IX, Town of Manapouri,	Southland Land District  Lot 1, D.P. 10441, situated in Block I, Manapouri Survey District.  Section 1, S.O. Plan 11736, situated in Block I, Manapouri Survey District.  Section 2, S.O. Plan 11736, situated in Block I, Manapouri Survey District.  Part Section 114, Block IX, Town of Manapouri, shown on S.O. Plan 5980.  Section 116, Block IX, Town of Manapouri, shown on S.O. Plan 9043.  Section 118, Block IX, Town of Manapouri, shown on S.O. Plan 9043.  Section 119, Block IX, Town of Manapouri, September 1991  Section 119, Block IX, Town of Manapouri, September 1991

### **Regulation Summary**

#### Notice Under the Acts and Regulations Publication Act 1989

Pursuant to the Acts and Regulations Publication Act 1989, notice is hereby given of the making of regulations as under:

Authority for Enactment	Title or Subject-matter	Serial Number	Date of Enactment	Price Code	Postage and Pack- aging
Diplomatic Privileges and	Diplomatic Privileges (European Bank for	1991/148	12/8/91	3-BX	\$2.10
Immunities Act 1968	Reconstruction and Development) Order 1991				
Conservation Act 1987	Fisheries (West Coast Whitebait Fishing)	1991/149	12/8/91	2-A	\$1.60
	Regulations 1985, Amendment No. 2				

#### Postage and Packaging Charge: Mail Orders

If two or more copies ordered, the remittance should cover the cash price and the maximum charge for the total value of purchases as follows:

Total Value of Purchases \$	Maximum Charge \$
\$12.00 and less	1.50
\$12.01 and greater	3.25

Copies can be bought or ordered by mail from GP Books Limited. Please quote title and serial numbers. Prices for quantities supplied on application.

GP Books Limited bookshops are located at Housing Corporation Building, 25 Rutland Street (Box 5513, Wellesley Street), Auckland 1; 33 Kings Street, Frankton (P.O. Box 857), Hamilton; Head Office, Mulgrave Street (Private Bag), Wellington 1; 25–27 Mercer Street (Private Bag), Wellington 1; 147 Hereford Street (Private Bag), Christchurch 1; Cargill House, 123 Princes Street (P.O. Box 1104), Dunedin; Shop 3, 279 Cuba Street (P.O. Box 1214), Palmerston North. ps8065

